Author's response to reviews

Title: Characterization of antibodies in single-chain format against the E7 oncoprotein of the Human papillomavirus type 16 and their improvement by mutagenesis

Authors:

Maria Gabriella Dona (dona@iss.it)
Colomba Giorgi (giorgi@iss.it)
Luisa Accardi (accardi@iss.it)

Version: 3 Date: 31 January 2007

Dear BioMed Central Editorial Production Team
the manuscript has been reformatted as requested according to the following:
1)Now the list of authors in the manuscript exactly matches that in the submission system. The mistake was concerning the first author, whose exact first name is "Maria Gabriella", and neither Maria G nor Gabriella M.
2)The Keywords section has been removed as requested.
Thank you very much for publication in BMC cancer.
Best regards
Luisa Accardi